Author’s response to reviews

Title: Reasons for late presentation for antenatal care, healthcare providers’ perspective

Authors:

Nelly Jinga (njinga@heroza.org;nelly.jinga@gmail.com)
Constance Mongwenyana (cmongwenyana@heroza.org)
Aneesa Moolla (amoolla@heroza.org)
Given Malete (gmalete@gmail.com)
Dorina Onoya (donoya@heroza.org)

Version: 4 Date: 17 Dec 2019

Author’s response to reviews:

Editor Comments:

1. Consent for publication
We note that you have not included a ‘Consent for publication’ section in Declarations. If identifying images or other personal or clinical details of participants are presented that compromise anonymity, a statement of consent to publish from the patient should be included. This section must be included even if it is not applicable to your manuscript. If consent to publish is not applicable to your manuscript, please write ‘Not Applicable’ in this section.
• Done – line 442

2. All authors read and approved
Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.
• Done – lines 460-61

3. Clean manuscript
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strike-throughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
• Done – One table and one figure uploaded separately